

Citation: Accoti A, Multini LC, Diouf B, Becker M, Vulcan J, Sylla M, et al. (2023) The influence of the larval microbiome on susceptibility to Zika virus is mosquito genotype-dependent. PLoS Pathog 19(10): e1011727. https://doi.org/10.1371/journal. ppat.1011727

Editor: Elizabeth A. McGraw, Pennsylvania State University - Main Campus: The Pennsylvania State University - University Park Campus, UNITED STATES

Received: June 2, 2023

Accepted: September 29, 2023

Published: October 30, 2023

Copyright: © 2023 Accoti et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Raw sequencing data is available through accession number PRJNA1027868.

Funding: LBD, AG, and MD were supported by U01AI151801 West African Center for Emerging Infectious Diseases. MB was supported by a T32 predoctoral fellowship (NIAID Emerging and Tropical Infectious Diseases Training Program AI007526, https://www.utmb.edu/cbeid). The RESEARCH ARTICLE

The influence of the larval microbiome on susceptibility to Zika virus is mosquito genotype-dependent

Anastasia Accoti¹, Laura C. Multini¹, Babakar Diouf², Margaret Becker^{1,3}, Julia Vulcan¹, Massamba Sylla⁴, Dianne Y. Yap⁵, Kamil Khanipov⁵, Mawlouth Diallo^{2,3}, Alioune Gaye^{2,3}, Laura B. Dickson^{1,3,6,7}*

1 Department of Microbiology and Immunology, University of Texas Medical Branch, Galveston, Texas, United States of America, 2 Medical Zoology Unit, Institute Pasteur Dakar, Dakar, Senegal, 3 West African Center for Emerging Infectious Diseases, Centers for Research in Emerging Infectious Diseases, Galveston, Texas, United States of America, 4 Laboratory Vectors & Parasites, Department of Livestock Sciences and Techniques Sine Saloum University El Hadji Ibrahima NIASS (USSEIN), Kaffrine, Senegal, 5 Department of Pharmacology and Toxicology, University of Texas Medical Branch, Galveston, Texas, United States of America, 6 Institute for Human Infections and Immunity, University of Texas Medical Branch, Galveston, Texas, United States of America, 7 Center for Vector-borne and Zoonotic Diseases, University of Texas Medical Branch, Galveston, Texas, United States of America

* Idickson@utmb.edu

Abstract

The microbiome of the mosquito Aedes aegypti is largely determined by the environment and influences mosquito susceptibility for arthropod-borne viruses (arboviruses). Larval interactions with different bacteria can have carry-over effects on adult Ae. aegypti replication of arboviruses, but little is known about the role that mosquito host genetics play in determining how larval-bacterial interactions shape Ae aegypti susceptibility to arboviruses. To address this question, we isolated single bacterial isolates and complex microbiomes from Ae. aegypti larvae from various field sites in Senegal. Either single bacterial isolates or complex microbiomes were added to two different genetic backgrounds of Ae. aegypti in a gnotobiotic larval system. Using 16S amplicon sequencing we showed that the bacterial community structure differs between the two genotypes of Ae. aegypti when given identical microbiomes, and the abundance of single bacterial taxa differed between Ae. aegypti genotypes. Using single bacterial isolates or the entire preserved complex microbiome, we tested the ability of specific larval microbiomes to drive differences in infection rates for Zika virus in different genetic backgrounds of Ae. aegypti. We observed that the proportion of Zika virus-infected adults was dependent on the interaction between the larval microbiome and Ae. aegypti host genetics. By using the larval microbiome as a component of the environment, these results demonstrate that interactions between the Ae. aegypti genotype and its environment can influence Zika virus infection. As Ae. aegypti expands and adapts to new environments under climate change, an understanding of how different genotypes interact with the same environment will be crucial for implementing arbovirus transmission control strategies.

funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Adult mosquitoes transmit many viruses, including Zika virus, during the process of taking a bloodmeal from human hosts. An important parameter of how well a mosquito is at transmitting viruses is whether the mosquito can become infected and replicate the virus. Different mosquito populations can be genetically distinct from each other, and in some cases genetic differences are associated with the habitat they live in. An important factor of the mosquito habitat is the water source in which larvae develop into adults. These water sources harbor diverse bacterial communities and are the source of the larval microbiome, which is known to influence how well the mosquito can become infected with viruses as an adult. Here we show that the effect of the larval microbiome on adult susceptibility to Zika virus depends on the genotype of the mosquito. These results indicate that different genetic backgrounds of mosquitoes interact with their habitat differently and this has important consequences for how easy it is for a mosquito to become infected.

Introduction

Arthropod-borne viruses (arboviruses) transmitted by mosquitoes represent a major cause of morbidity and mortality worldwide [1,2]. The mosquito, *Aedes aegypti*, is the main vector for arboviruses worldwide including dengue (DENV), Zika (ZIKV), yellow fever (YFV), and chi-kungunya (CHIKV) viruses. Climate change and a warming world exacerbate this risk from vector borne diseases [3,4] by expanding the range of vectors. Additionally, arbovirus epidemics are poised to be a major threat in sub-Saharan Africa [5]. Given the ongoing and increasing risk of mosquito-borne viruses, especially in Africa, it is crucial to understand factors that contribute to their emergence and transmission.

Aedes aegypti demonstrates large phenotypic variability in its interactions with arboviruses, largely driven by genetic and environmental variation. *Aedes aegypti* is genetically diverse worldwide [6–8] where most of the genetic diversity is observed within Africa. In West Africa, genetic variation between populations of *Ae. aegypti* is largely driven by degree of urbanization and rainfall and linked to adaptation to human preference [7] and its ability to efficiently transmit arboviruses is a partially genetically controlled trait [9]. Specifically, different genotypes of *Ae. aegypti* [10–12] have been shown to result in different infection outcomes with arboviruses. In some cases, vector competence is dependent on the specific mosquito-virus interactions [13–15].

Additionally, abiotic (non-living) [16–29] and biotic (living) environmental factors [30–36] are known to contribute to the vector competence of *Ae. aegypti*. An important biotic ecological parameter influenced by the environment is the larval microbiome. Globally, *Ae. aegypti* occupies a variety of environments and diverse larval habitats. Outside Africa, domesticated *Ae. aegypti* oviposits in artificial containers such as discarded buckets or cans and tires around human habitats. In Africa, *Ae. aegypti* will oviposit and develop in a variety of container types ranging from artificial containers around human habitats, to tree holes and rockpools in forested habitats. Larval development sites represent different microbiomes [37]. The larval microbiome is largely determined by the aquatic environment and is critical for establishing the nutritional status of the mosquito [38]. Importantly, interactions between *Ae. aegypti* larvae and different bacterial strains have carry-over effects that drive variation in DENV susceptibility in adults [37,38]. However, whether different genotypes of *Ae. aegypti* interact differently with the same larval microbiome to drive variation in arbovirus susceptibility

remains unknown. As the earth becomes warmer and drier and de-forestation and urbanization increase, *Ae. aegypti* may expand into new environments [39,40] and exploit different oviposition container types. This is especially true in Africa where the larval development site is tied to the environment. *Aedes aegypti* genotypes accustomed to ovipositing in forest or natural larval sites may be forced to adapt and oviposit in urban artificial container types.

Here we expand on previous work [37] to determine if the carry-over effects of the larval microbiome on arbovirus susceptibility is dependent on mosquito genotype. Using single bacterial isolates, we demonstrate the influence of specific bacterial isolates on adult *Ae. aegypti* susceptibility to ZIKV is dependent on mosquito genotype. Using complex microbiomes isolated from larvae in Senegal, we demonstrate that different genotypes of *Ae. aegypti* retain different members of these larval microbiomes and that ZIKV susceptibility is dependent on the specific pairing between *Ae. aegypti* genotype and complex microbiome during larval development. Our results provide empirical evidence that *Ae. aegypti* genotype by microbiome interactions drive variation in ZIKV susceptibility.

Results

Single bacterial isolates

Previously we observed that larval exposure to different individual bacterial isolates has carryover effects which alter DENV replication in Ae. aegypti [37], but it remains unknown if this influence is consistent across vector genotypes. To address this, we measured whether a single bacterial isolate has the same carry-over effects on ZIKV infection between different genotypes of Ae. aegypti using a gnotobiotic assay. Of the 27 bacteria isolated from larvae collected from large metal drums in Thiés, Senegal, three bacterial isolates were chosen (Serratia spp. (Bacterial Isolate A), Chyrseobacterium spp. (Bacterial Isolate B), and Serratia spp. (Bacterial Isolate C)) for further characterization (see Material and Methods for information on selection). To understand if larval exposure to specific bacterial isolates alters pupation rates in an Ae. aegypti genotype-dependent manner, two lines of Ae. aegypti from Senegal with known genetic differences [7], Thiés (THI) and Kedougou (KED), were exposed to equal numbers of bacteria from each isolate in a gnotobiotic system, and the proportion of larvae that had pupated was recorded each day. Pupation rates were measured in triplicate gnotobiotic flasks daily from the onset of pupation (day five post-hatching) until day 10. Overall, the THI line of Ae. aegypti pupated more slowly than the KED line (p. value < 0.0001). In both lines, the time to reach 50% pupation (PD₅₀) was different among the three bacterial isolates (Fig 1A) (S1 Table). In both mosquito lines the bacterial isolate associated with the slowest pupation rate was Bacterial Isolate A with a PD₅₀ of 8.5 days in the THI line and 7.4 days in KED line. Conversely, the bacterial isolate that resulted in quickest pupation was Bacterial Isolate B where a PD_{50} of 6.9 days occurred in the THI line and 6.1 days in the KED line. Bacteria C resulted in intermediate pupation rate of 7.4 days in the THI line and 6.7 days in the KED line. Interestingly, Bacteria A and C both belong to the genus Serratia but exert different effects on pupation rates. Overall, we observed that the PD_{50} was dependent on the mosquito genotype, bacterial isolate, or the interaction between the two variables (ANOVA on a general linearized model: mosquito genotype: p < 0.0001, bacterial isolate: p < 0.0001, mosquito genotype x bacterial isolate: p < 0.0001).

To determine if larval exposure to different single isolates influences ZIKV infection rates in a mosquito genotype dependent manner, the THI and KED line were challenged with ZIKV following larval exposure to Bacteria A, Bacteria B, or Bacteria C. Seven days post exposure to ZIKV, infection rates were determined by detection of viral RNA by RT-PCR. Infection rates ranged from 20–50% and was not dependent on the bacterial isolate (p-value = 0.614) or the





mosquito genotype (p-value = 0.372), but was dependent on the interaction between the bacterial isolate and the mosquito genotype (p-value = 0.010) (Fig 1B).

Complex microbiomes

Although using single bacterial isolates in a gnotobiotic system is more tractable, it is not representative of natural mosquito microbiomes. To establish if the effect of the interaction between mosquito genotype and bacterial isolate on susceptibility to Zika virus infection can be extended to complex microbiomes collected from larvae in the field, whole microbiomes from the larvae pools collected from five sites in Thiés, Senegal were introduced to both genotypes of Ae. aegypti. To establish if we were introducing different microbiomes to axenic larvae, we characterized the microbiomes of the larvae naturally developing in the collection sites at the same time of collection of larvae used for the gnotobiotic assay by 16S amplicon sequencing. Principal component analysis (PCA) on a Bray-Curtis dissimilarity matrices demonstrate that the initial bacterial community structure was different between sites (S1 Fig). We then characterized the bacterial communities of the gnotobiotic larvae following inoculation with the larval homogenates from each site to confirm the five microbial communities remained different. Out of 80 individual larvae sequenced, a total of 87 OTUs were found, which represent 30 genera after filtering for low abundance OTUs and OTUs present in the negative controls. Rarefaction curves (S2 Fig) showed that sufficient sequencing depth was achieved. Principal component analysis (PCA) was performed on a Bray-Curtis dissimilarity matrix on the larvae from the same sites to determine if the bacterial communities from each



Fig 2. The bacterial community structure differs between the gnotobiotic larvae. Structure of bacterial communities of KED and THI larvae combined was determined by deep sequencing the V3-V4 region of the 16S gene in individual larvae reared in a gnotobiotic system inoculated with complex microbiomes generated from larvae homogenates collected in Senegal (Site1-Site5). Bacterial structure is represented by PCoA of a Bray-Curtis dissimilarity matrix based on mean genera abundance (PERMANOVA p = 0.001).

collection site that were introduced to the larvae were in fact different. The community structure differed among all five of the bacterial communities (Fig 2) (PERMANOVA, p-value = 0.001). The bacterial community from Site 1 was the most different from the other sites, and Sites 2–5 showed greater similarity.

To assess whether the two genotypes of *Ae. aegypti* acquired and maintained different bacterial taxa after larval exposure to an identical bacterial community in a gnotobiotic system, the percent abundance of different genera was compared among genotypes and collection sites. The abundance of the top 20 most prevalent genera differed in both genotypes between the collection sites (Fig 3A). Gnotobiotic larvae harboring the bacterial community from Sites four or five were most similar between the genotypes. Gnotobiotic larvae harboring the bacterial community from Sites one, two, and three were most different between the genotypes. Abundance of specific genera was consistent among individuals from each treatment (S3 Fig). To determine if there are specific genera that are differentially abundant between genotypes in



Fig 3. Abundance of specific taxa and community structure differs between lines of gnotobiotic larvae. (A) The percent abundance of the top 20 most abundant genera is plotted by larval treatment (Site1-5) and mosquito line (KED or THI). (B) Beta diversity metrics for the *Ae. aegypti* lines in each bacterial treatment analyzed separately. The dissimilarities between the two different lines of *Ae. aegypti* (KED and THI) in each of the five different larval microbiomes was analyzed by principal component analysis of Bray-Curtis dissimilarity matrixes. P-values from individual PERMANOVA analysis are as follows Site 1: p = 0.002, Site 2: p = 0.002, Site 3: p = 0.001, Site 4: p = 0.058, Site 5: p = 0.005).

the gnotobiotic system, pairwise differential abundances of each genus were compared between mosquito genotypes after exposure to each of the five complex microbiomes. Of the 30 genera identified in this study, five to seventeen genera were differentially abundant between genotypes. (S2 Table). Additionally, the overall bacterial community structure differed between mosquito genotypes when fed identical microbiomes. When all sites were analyzed together, the overall bacterial community structure differed between sites and genotypes (PERMANOVA p = 0.001) (S4 Fig). When each microbiome source was analyzed independently, the community structure differed between mosquito genotypes following exposure to microbiomes from Sites 1, 2, and 3, but not between Sites 4 or 5 (Fig 3B).

To evaluate if the mosquito genotype by larval microbiome interactions on adult Ae. aegypti susceptibility to arboviruses extends to complex microbiomes, the two different mosquito genotypes were each exposed to the same preserved complex microbiomes harvested from larvae from five natural Senegal habitats and challenged with ZIKV. Ingestion of the homogenized larvae from the five sites results in different bacterial community structures in the larvae (Fig 2), indicating that the mosquitoes were exposed to five different complex microbiomes during larval development prior to be being challenged with ZIKV as an adult. The proportion of infected individuals was determined seven days post oral exposure by detecting viral RNA by RT-PCR. The proportion of infected mosquito bodies varied based on the larval microbiome it was exposed to and the mosquito genotype. Specifically, the proportion of the infected bodies was not dependent on the bacterial community (p-value = 0.265) or the mosquito genotype (p-value = 0.392), but was dependent on the specific pairing of larval microbiome and genotype (p-value = 0.017) (Fig 4A). To determine if this infection phenotype extends to dissemination titers, the number of infectious particles was enumerated in the heads, a poxy for saliva [41], seven days post infection by focus forming assay. The titer of ZIKV in the heads was dependent on the bacteria community (p-value = 0.011), but not on the mosquito genotype (p-value = 0.159) or on the interaction between mosquito genotype by larval microbiome interaction (p-value = 0.772) (Fig 4B).

Discussion

In this study, we explored the contribution of mosquito genotype and larval microbiome in driving variation in ZIKV susceptibility. Having previously observed that adult replication of DENV is dependent on the specific bacteria that the mosquito was exposed to during larval development [37], we sought to determine if the effect of larval gnotobiotic treatment on arbovirus susceptibility was mosquito genotype-dependent. We found that the proportion of ZIKV-infected adults is dependent on larval exposure to individual bacterial isolates during larval development and *Ae. aegypti* genotype. When exposed to identical complex microbiomes in a gnotobiotic system, the different mosquito genotypes differed in the abundance specific genera and in the bacterial community structure retained. Finally, we observed that the proportion of ZIKV-infected bodies, but not the head titers, was dependent on the specific pairing between larval microbiome. Together these data demonstrate that different genotypes of *Ae. aegypti* interact differently with their larval microbiomes and these genotype-dependent interactions have carry-over effects important for ZIKV susceptibility.

In accordance with previously published work [37], we did not see an effect of larval exposure to a single bacterial isolates on the proportion of ZIKV-infected heads. Dickson et al. 2017 [37], only observed an influence of larval exposure to different bacteria on the DENV infectious load in the head, perhaps related to innate immune activation controlling viral replication and dissemination. Here, we did not assay dissemination titers. If done, we might also observe differences in the amount of infectious virus outside the midgut dependent on which bacterial isolate the larvae were exposed to. Nonetheless, we still detect an interaction between mosquito genotype and larval microbiome on the proportion of ZIKV-infected bodies.



Fig 4. Larval development in different complex microbiomes alters ZIKV infection rates in a mosquito genotype dependent manner. (A) The proportion of blood-fed *Aedes aegypti* females from the KED and THI lines with a ZIKV-positive body 7 days post-infectious blood meal by RT-PCR following larval rearing in complex microbiomes, Site1-5. The y-axis indicates the proportion of ZIKV-infected female bodies and error bars represent the 95% confidence intervals of the proportions. Data were analyzed by binomial logistic regression as a function of bacterial treatment, mosquito genotype, and their interaction (bacterial treatment: p-value = 0.265, mosquito genotype: p-value = 0.392, but was dependent on the specific pairing of bacterial treatment x mosquito genotype: p-value = 0.017. The number of individual mosquitoes is shown within each bar. (B) Boxplot showing the dissemination titers of infectious ZIKV particles expressed as the Log10-transformed number of focus-forming units (FFU) per ml detected in the *Ae. aegypti* head seven days post-infectious blood meal. The points represent individuals and the mean is represented by a horizontal line. The error bars represent the 95% confidence interval. Data were analyzed by Two-way ANOVA as a function of bacterial treatment, mosquito genotype; p-value = 0.11, mosquito genotype: p-value = 0.159, bacterial treatment x mosquito genotype, and their interaction (bacterial treatment: p-value = 0.011, mosquito genotype: p-value = 0.159, bacterial treatment x mosquito genotype: p-value = 0.772). The number of individual mosquitoes assayed is Site 1 Ked: 5, Site 1 THI: 8, Site 2 Ked: 7, Site 2 THI: 8, Site 3 Ked: 5, Site 3 THI: 6, Site 5 Ked: 3, Site 5 THI: 3.

Interestingly, we did observe that ZIKV infectious loads in the mosquito head was dependent on complex bacterial community that the larvae were exposed to in accordance with Dickson et al. 2017 [37], but no interaction between mosquito genotype and larval microbiome was detected. This suggests that mosquito genotype by larval microbiome interactions only have carry-over effects on infection rates, while the influence of the larval microbiome on the amount of disseminated virus is largely driven by the larval microbiome across mosquito genotypes, perhaps due to conserved differences in specific immune responses across genotypes. The mosquito immune system is known to respond to different microbiomes [32,42]. Furthermore, the results observed in this study could be dependent on the ZIKV isolate used, which originated from Senegal and produces higher infection and dissemination rates than epidemic isolates of ZIKV [43]. Perhaps we could detect an interaction between mosquito genotype and larval microbiome on dissemination titers if we had used a different isolate of ZIKV.

The nutrition status of the mosquito and larval microbiome have previously been shown to influence mosquito fitness [44,45] and susceptibility to arboviruses [37,46–48]. The microbiome is composed of diverse microorganisms that colonize the mosquito's gut, interacting

with the host's metabolic processes [46,49,50] and modulating its innate immune response [36,51–53]. Additionally, recent research has shed light on the critical role of nutrition in determining the outcome of infection in mosquitoes. In particular, lipid metabolism plays a crucial role in the replication and dissemination of arboviruses within the mosquito's body [54–59], while amino acids are involved in the mosquito's interaction with the microbiome [49,60]. Given that the mosquito's nutritional status is strongly influences by its microbiome, and that the nutritional status of the mosquito can influence arbovirus infection, it is not surprising that we observed significant variation in the rates of ZIKV infection when mosquitoes were reared in different bacterial communities.

Additionally, there have been several mosquito genes identified that impact the microbiome composition and gut equilibrium [49,61–64]. These genes can control the overall abundance of the microbiome or specific taxa, through their involvement in bloodmeal digestion and immune factors [61,64]. Moreover, the mosquito microbiome has the potential to influence the expression of particular genes, which can shape the mosquito's immune response and facilitate efficient colonization of specific microbes [63]. Although it remains uncertain which mosquito genes are responsible for the observed phenotypes in this study, it is possible that genetic variation in genes regulating specific bacterial taxa exist between the two genotypes of *Ae. aegypti* utilized in this study.

While numerous studies have identified mosquito genes that interact with the microbiome [49,61–64] or have identified bacteria that influence mosquito interactions with arboviruses [32,36,65–71], to the best of our knowledge there are no studies which directly test the interaction between the mosquito larval microbiome and mosquito genotype on arboviruses infection outcome. The importance of host genotype by microbiome interactions on various phenotypes has been demonstrated in other organisms such as *Drosophila* [72,73] and bumble bees (*Bombus terrestris*) [74]. Given the importance of mosquito genotype [75–78] and the importance of the mosquito microbiome arboviruses [32,36,65–71] in driving interactions with pathogens, it is relevant to investigate how these two variable interact to drive variation in mosquito-pathogen interactions.

An important finding of this study is that the different genotypes of Ae. aegypti larvae harbor different abundances of specific taxa after being fed identical microbiomes. Multiple studies have sought to determine if different genotypes or lines of mosquitoes have the same microbiome when maintained in the same environment. While some studies observed no differences in the microbiome between genotypes of Ae. aegypti [79], others observed different microbiomes between lines in the same environment [80] and these changes held up across microbially diverse environments [81]. While it is assumed that the bacteria the mosquitoes are exposed to in the same insectary is the same across lines, this is not absolute and it is very plausible that, during rearing, some larval trays could contain different microbes. By using a gnotobiotic system, we ensured that the different genotypes received the same microbiome in a highly controlled environment. We observed differences in the abundance of specific genera between different genetic backgrounds when fed identical microbiomes. This demonstrates that the mosquito genetic background plays a role in microbiome composition. Other factors that contribute to microbiome composition are the environmentally available bacteria [50], and competition between bacteria [82]. Given that the microbiome is remodeled during the transition from larvae to adult development, it is likely that that adult microbiomes are different than what was provided to larvae, so we cannot conclude that the adult microbiome was different between the two genotypes. Nevertheless, our work further highlights the importance of mosquito genotype or mosquito line in shaping the microbiome.

Even though these microbiomes originated from larvae collected in Senegal, we cannot make any conclusions regarding their field relevance given that we likely partially altered their composition through preservation and transport compared to larvae in the original habitat. However, how well these microbiomes recapitulated those in nature is not relevant for this study. We simply aimed to show that different complex bacterial communities can impact ZIKV infection in a mosquito genotype-dependent manner, and using microbiomes harvested from larvae in the field is more relevant than ad hoc mixing single bacterial isolates, even if our complex bacterial communities are not identical to those in the field. Perhaps if we seeded our gnotobiotic flasks with the microbiomes from more larvae in accordance with recent studies showing you can transplant and preserve the microbiome [83], we could make conclusions based on the origin of microbiomes.

Two of the single bacterial isolates used in this study (Bacteria A and Bacteria C) belong to the same genus, yet we observed differences in the pupation rate and infection rates between these two bacteria in a mosquito population dependent manner. The influence of these isolates on pupation rate was consistent across mosquito genotypes, but the influence of these isolates on ZIKV infection was the opposite direction across mosquito genotypes. In fact, the influence of these two isolates is likely driving the observed interaction between bacterial isolate and mosquito genotype. Perhaps genetic differences between these closely related bacteria species have variable interactions with different mosquito genotypes which are important for ZIKV infection. This system could provide a highly tractable system to investigate the mechanism of larval microbiome by mosquito interactions on arboviruses susceptibility.

Taken together our results show that different genotypes of *Ae. aegypti* interact with their larval microbiome differently to influence ZIKV infection. Future studies should expand on this work to mechanistically identify how different microbiomes influence infection outcomes of the mosquito in a mosquito genotype-dependent manner.

Materials and methods

Bacterial isolation

Mosquito larvae were collected from five sites in two locations (Dixième and Keur Dabo Ndione) in Thiès, Senegal. Thiès is a city in the Northwestern part of Senegal. All larvae were collected from large plastic drums during the dry season of 2021. A pool of 3 larvae from each site was rinsed in sterile 1X PBS (Phosphate-buffered saline), incubated in 70% Ethanol for 5 minutes (min), and then rinsed in sterile 1X PBS three times. Next, they were homogenized in 500 µl sterile 1X PBS and 30% glycerol was added.

A portion of the glycerol stock from Site one containing homogenized larvae was plated on Trypticase Soy Agar (rich media) and incubated for 3 days at 30°C. Individual colonies were picked from the plates and used to inoculate 3 ml of LB media, which were shaken at 30°C until bacterial growth occurred (1 OD) and used to create new glycerol stocks of the individual isolates. DNA was extracted from each colony with the QIAGEN Dnaesy blood and tissue kit following the manufacture's protocol. The bacterial DNA was used to amplify the entire 16S region by PCR [5'-AGAGTTTGATCCTGGCTCAG-3' (forward) and 5'-AAGGAGGTGATC-CAGCCGCA-3' (reverse)] using Expand High-Fidelity Polymerase (Sigma-Aldrich). The PCR products were purified using the QIAquick PCR Purification kit (Qiagen), quantified by NanoDrop (NanoDrop Technologies Inc.), and sequenced by Sanger sequencing (Molecular Genetics Facility at University of Texas Medical Branch). The sequences were aligned and classified at the genus level using the SILVA database (www.arb-silva.de/). A total of 27 isolates were isolated from Site one and 22 were identified taxonomically from (S3 Table). This list does not represent the entire population of bacteria that was isolated. The intended purpose of isolating bacteria from these samples was to have isolates from Ae. aegypti that colonize the larvae in the field for use in our gnotobiotic assay. We were not intending to fully characterize

the culturable bacteria from each site. Three isolates *Serratia spp*. (Bacterial Isolate A), *Chryseobacterium spp*. (Bacterial Isolate B), and *Serratia spp*. (Bacterial Isolate C). *Chryseobacterium* was chosen for the gnotobiotic assay based off their presence in the larvae after larvae development in the bacterial community from Site One (Fig 3), indicating it was good at colonizing the larvae. The *Serratia* isolates were chosen given previous associations of blocking pathogens in mosquitoes [84,85]. These isolates were not chosen to test any specific hypothesis about the isolates, draw conclusions about the isolates themselves, or to mirror any field relevance.

To standardize the amount of bacteria that would be introduced into the gnotobiotic system, aliquots of equal amounts of bacteria were made. The amount of aliquoted bacteria was then quantified by enumerating colony forming unit (CFU) for each bacterial isolate. To make the aliquots, 200 μ l of each bacterial glycerol was added to 200 ml of LB and shaken at 30 °C until bacterial growth occurred, then 50 ml was pelleted by 3000 rpm centrifugation for 15 min. The pellet was washed two times with 50 ml fresh LB broth. After the second wash, the pellet was resuspended in 50 ml of LB and aliquots were made by mixing 500 μ l of resuspended bacteria in each stock, 10 μ l was taken from an aliquot and serially diluted and plated on LB plates. The number of colonies were counted and used to calculate CFU/ml.

Gnotobiotic larvae

To create axenic larvae, Aedes aegypti eggs were collected from seventh-generation and eighthgeneration laboratory colonies of Thiés (THI) and Kédougou (KED), respectively, derived from natural populations from Thiés, Senegal, and Kédougou, Senegal [7]. Colonies were made by collecting eggs from each colony using ovitraps as described in [7]. Thiés is located in the Northwest part of Senegal and Kedougou is located in the Southeast part of the Senegal. Mosquitoes from these two locations differ in preference for humans and these two locations differ in the degree of urbanization, and the amount of rainfall and genomic data for these lines exist and they represent different genotypes of Ae. aegypti [7]. Eggs were gently scrapped off the paper into a 50 ml falcon tube. The eggs were sterilized by incubation in 70% ethanol for 5 min, 3% bleach for 3 min, and 70% ethanol for 5 min. The eggs were then rinsed in distillated (d) sterile water three times and then they were allowed to hatch in 30 ml of d-water in a 50 ml falcon tube with a $0.2 \,\mu$ M filter lid. Upon hatching, as a control, 10–15 axenic larvae were transferred to a sterile 25 cm^2 tissue-culture flask containing 15 ml of d-water and 50 μ l of sterile fish food (1 gram ground fish food flakes per 10 ml d water autoclaved for 20 minutes at 121°C). These axenic larvae were used as an egg-sterilization control and did not develop past the 1st instar larval stage in accordance with previously published work [86]. Gnotobiotic larvae were made by distributing 50 ± 5 (T-75 cm² tissue-culture flasks) or 80 ± 20 (T-150 cm² tissue-culture flasks) axenic larvae to sterile 75 or 150 cm² tissue-culture flasks in duplicate or triplicate containing either 45 or 120 ml of d-sterile water and 1 ml of sterile fish food. For the single isolate, 5 x10⁵ CFUs/ml of washed-bacteria (for details go to bacteria growth section) was added to each flask. For each bacterial isolate tested, three replicate flasks were used. A total of two independent experiments was performed. To make gnotobiotic larvae with complex microbiomes, equal amounts of a single glycerol stocks from each collection site (Sites 1-5) were added to each of two duplicate T-150 flasks. By adding a homogeneous mixture from a single tube, each flask and mosquito genotype is receiving the same bacterial inoculum. Data represents one experimental replicate due to availability of field material. Control and gnotobiotic larvae were maintained on 50 (T-75 flask) and 500 (T-150 flask) µl sterile fish food every other day, respectively. Bacterial treatments were added immediately following hatching at the L1 stage. Following pupation and eclosion, adults were maintained under standard

insectary conditions and allowed to be colonized by environmental bacteria. This was done because we are measuring the carry-over effects of the larval microbiome and wanted the adult microbiome to be seeded under standard insectary conditions.

16S and metagenomic analysis

To characterize the microbiome of the larvae developing in the field sites prior to use in the gnotobiotic assay, larvae were collected and surface sterilized in 70% ethanol for 5 min and rinsed 3 times in sterile dwater before placing in RNALater (Qiagen) and frozen at -80°C and transported to UTMB. DNA was extracted by placing individual larvae into 2 ml tube containing a 5mm grinding bead and homogenized for 3 minutes at a 30Hz/s frequency in a TissueLyser II grinder (Qiagen). DNA extraction of individual larvae was carried out using the QIAamp DNA Kit (Qiagen, Germany) following the manufacturer's protocol. Larvae for initial 16S characterization and for use in the gnotobiotic assay were collected and processed at the same time.

To characterize the microbiome in the gnotobiotic larvae seeded with the complex microbiomes, eight individual L3 larvae were collected from each treatment flask and transferred to a 96-well cell culture plate. The larvae were surface sterilized in 70% ethanol for 5 min and rinsed 3 times in sterile dwater. Next, individual larvae were transferred to 2 ml tubes containing a 5mm grinding bead and placed in the -80°C freezer until DNA was extracted. Individual mosquitoes were homogenized for 3 minutes at a 30Hz/s frequency in a TissueLyser II grinder (Qiagen). DNA extraction of individual larvae was carried out using the QIAamp DNA Kit (Qiagen, Germany) following the manufacturer's protocol. No-mosquito controls were used for each extraction batch and included in the sequencing run.

Sequencing libraries for each isolate were generated using universal 16S rRNA V3-V4 region primers [87] in accordance with Illumina 16S rRNA metagenomic sequencing library protocols. DNA concentrations of each library were determined by Qubit and equal amounts of DNA from each barcoded library were pooled prior to sequencing. The samples were barcoded for multiplexing using Nextera XT Index Kit v2. The pooled libraries were diluted to 4 pM and run on the Illumina Miseq using a MiSeq Reagent Kit v2 (500-cycles).

To identify known bacteria, sequences were analyzed using the CLC Genomics Workbench 21.0.5 Microbial Genomics Module (CLC MGM). Reads containing nucleotides below the quality threshold of 0.05 (using the modified Richard Mott algorithm) and those with two or more unknown nucleotides or sequencing adapters were trimmed out. Reference-based Operational Taxonomic Unit (OTU) picking was performed using the SILVA SSU v132 97% database [88]. Sequences present in more than one copy but not clustered to the database were placed into de novo OTUs (97% similarity) and aligned against the reference database with an 80% similarity threshold to assign the "closest" taxonomical name where possible. Chimeras were removed from the dataset if the absolute crossover cost was three using a k-mer size of six. OTUs with a combined abundance of less than two were removed from the analysis. Low abundance OTUs were removed from the analysis if their combined abundance was below 10 or 0.1% of reads. The number of reads per sample used in the analysis ranged from 13,214–111,218. Only reads that mapped to bacteria were kept. Taxa classified as "Ambiguous Taxa" are reads mapping to bacterial DNA, but that cannot be identified at the taxonomic level.

Pairwise differential abundance of specific genera was done in MicrobiomeAnalyst. Statistical significance between groups was determined by T-test/ANOVA and corrected for multiple testing. Presented p-values reflect correction for multiple testing. (S2 Table).

Abundance profiling was performed using MicrobiomeAnalyst [89,90]. The analysis parameters were set so that OTUs had to have a count of at least 10 in 20% of the samples and

above 10% inter-quantile range. Analysis was performed using actual and total sum scale abundances. Alpha diversity was measured using the observed features to identify the community richness using Chao1. Statistical significance was calculated using T-test/ANOVA. Beta diversity was calculated using the Bray-Curtis dissimilarity measure (genus level). Permutational Multivariate Analysis of Variance (PERMANOVA) analysis was used to measure effect size and significance on beta diversity for grouping variables [91]. Relative abundance analysis was done in MicrobiomeAnalyst at the level of genera.

Out of 80 individual larvae sequenced, a total of 1768 OTUs were identified. After filtering, 92 OTUs remain which represent 30 genera. Sequences from three individuals were removed from the analysis because they did not achieve enough reads, one from Site 1, 4, and 5. After removing OTUs belonging to the negative control, 87 OTUs remained. These final 87 OTUs were used for the analysis in the MicrobiomeAnalyst.

Pupation rate

Pupae were counted from the onset of pupation (Day 5) until a majority of the larvae pupated (Day 10) in the same triplicate flasks used for the adult viral challenge assays. Larvae that did not pupate were counted and considered as total amount of individuals. To determine the rate of pupation, the percent pupae was determined at each day by dividing the number of pupae by the total number of individuals. Data from two independent experiments was used, each with three internal replicates (three replicate flasks). Graphpad (Version 8) was used to generate a simple logistic regression that computed the day that 50% of larvae pupated (PD₅₀). An ANOVA was run on the summary statistics of the PD₅₀ generated from the logistic regression to determine if the PD₅₀ was dependent on the bacterial isolate, the mosquito genotype, or an interaction between the two. Multiple comparison by-two-way-ANOVAs were performed to compare the mean PD₅₀ between populations (Sidak's test), and between each of three different bacterial treatments within population (Tukey's test) and between each bacterium in the two populations (Sidak's test). Data are a summary of two biological experiments done each time in triplicates. The number of larvae used per experiment along with statistical information associated with each comparison are listed in S1 Table.

Mosquito infections

Mosquito infection assays was conducted using the ZIKV DAKAR 41524 isolate received from the World Reference Center for Emerging Viruses and Arboviruses at UTMB. After pupation, pupae were transferred to a 1-pint carton box with netting and 10% sucrose solution until adult emergence. After adult emergence, 4 to 5-day-old females were sorted and transferred into a new cup with netting and deprived of sucrose solution for 24 hours and transferred to an Arthropod Containment level 2 facility (ACL-2). Females were offered an artificial blood meal for 15 minutes using the Hemotek system with de-salted pig intestine as the membrane. The infectious blood meal consisted of a 2:1 mixture of defibrinated sheep blood (Colorado Serum Company) and virus at a final concentration of 1.49×10^7 focus-forming units (FFU)/ml. The blood meal was supplemented with 10 mM adenosine triphosphate (ATP). Prior to addition to the blood, sodium bicarbonate was mixed with the virus stock at 1% final concentration. Following exposure to an infectious bloodmeal, fully engorged females were sorted into 1-pint carton boxes with ad libidum access to 10% sucrose solution and kept in an incubator under controlled conditions (28°C, 12h:12h light: dark cycle). After 7 days of incubation, the head and body of ZIKV-exposed mosquitoes were separated to determine infection rate (the proportion of blood-fed mosquitoes with ZIKV-positive body) and dissemination titer (the amount of virus in the head tissues of ZIKV-infected mosquitoes). To determine the infection rate, female

bodies were homogenized in 200 µl of a crude RNA extraction buffer (10 mM Tris HCl, 50 mM NaCl, 1.25 mM EDTA, fresh 0.35 g/L proteinase K) during two rounds of 3 minutes at a 30Hz/s frequency in a TissueLyser II grinder (Qiagen). Total RNA was converted into complementary DNA (cDNA) using M-MLV reverse transcriptase (Invitrogen) and random hexamers, the reaction was carried out as follows: 10 min at 25°C, 50 min at 37°C, and 15 min 70°C. The cDNA was amplified by PCR carried out in a 25µl reaction containing 12.5µl of 1x DreamTaq DNA polymerase (Thermo Fisher Scientific) and 10 µM of each ZIKV primer (forward: 5'-GTATGGAATGGAGATAAGGCCCA-3', and reverse: 5'-ACCAGCACTGCCATTGATG TGC-3'). Cycling conditions were as follow, 2 min at 95°C, followed by 35 cycles of 30s at 95°C, 30s at 60°C, and 30s at 72°C with a final extension step of 7 min at 72°C. Amplicons were visualized on a 2% agarose gel. The proportion of ZIKV-infected females was analyzed by binomial logistic regression as a function of treatment, colony, and their interaction in R.

To determine the dissemination titer, the heads of females with positive ZIKV-infected bodies were titrated by focus-forming assay in Vero cells. Only three sites were used due to low sample sizes in the other treatments. Heads were homogenized individually in 200 μ l of Vero cell media (DMEM 1X) supplemented with 2% heat inactivated fetal bovine serum (FBS) and 1X Antibiotic-Antimycotic (Life Technologies) for 3 minutes at a 30Hz/s frequency in a TissueLyser II grinder (Qiagen). Vero cells were seeded in 24-well plates and incubated for 24 hours to reach confluency. Each well was inoculated with 200 µl of head homogenate in 10-fold dilutions (from 10^1 to 10^6) and incubated at $37^{\circ}C$ (5% CO²) for 1 hour, rocking every 15 minutes. Infected cells were overlaid with α-MEM media supplemented with 1.25% carboxymethyl cellulose, 5% FBS, and 1% Pen-Strep. After three days of incubation at 37°C, infected cells were fixed with 10% formalin for at least 1 hour and cells were washed three times in 1X PBS. Approximately 500 µl of blocking solution (5% w/v non-fat powdered milk in 1X PBS) was added to each well and the plates were placed on the plate rocker for 30 minutes. The blocking solution was discarded and 200 µl of primary antibody (obtained from the World Reference Center for Emerging Viruses and Arboviruses (WRCEVA) at UTMB) (ZIKV antibody diluted 1:1000 in blocking solution) was added to each well and plates were placed on plate rocker overnight. The primary antibody solution was discarded, and plates were washed three times with 1X PBS again, and 200 µl of secondary antibody (peroxidase-labeled goat anti-mouse IgG human serum KPL-474-1806) solution (secondary antibody diluted 1:2000 in blocking solution) was added to each well. Plates were placed on plate rocker for 1 hour. The secondary antibody solution was discarded, and plates were washed three times with 1X PBS. To develop visible foci, 100µl of TrueBlue peroxidase substrate (KPL 5510-0050) was added to each well, and plates were placed on the plate rocker until foci could be seen, around 10 min. Plates were washed with deionized water and FFU was counted with the help of a light. Focusforming units were Log₁₀ transformed to represent the concentration of infectious ZIKV particles detected in Ae. aegypti heads. Head titer data were analyzed by two-way ANOVA as a function of bacterial treatment, mosquito genotype, and their interaction in R. Infection rate data was analyzed by a two-way ANOVA on a binomial logistic regress in R.

Supporting information

S1 Fig. The bacterial community structure differs between the larvae in the field collection sites. Structure of bacterial communities was determined by deep sequencing the V3-V4 region of the 16S gene in individual larvae collected from large metal drums at five sites in Senegal (Site1-Site5). Bacterial structure is represented by PCoA of a Bray-Curtis dissimilarity matrix based on mean genera abundance (PERMANOVA p = 0.001). (TIF)

S2 Fig. Rarefaction curves showing the sequencing depth of each library. The number of species is shown on the Y axis, and the number of sequencing reads is shown on the X axis. (TIF)

S3 Fig. The percent abundance of the top 20 most abundant genera is plotted by larval treatment (Site1-5) and mosquito line (KED or THI) and separated out by individual. (TIF)

S4 Fig. Beta diversity metrics for the *Ae. aegypti* lines in each bacterial treatment. The dissimilarities between the two different lines of *Ae. aegypti* (KED and THI) in each of the five different larval microbiomes was analyzed by principal component analysis of Bray-Curtis dissimilarity index (PERMANOVA, p = 0.001). (TIF)

S1 Table. Summary of statistics for pupation rate. (XLSX)

S2 Table. The log₂fold change of bacteria genera with significant pairwise differences between KED and THI is plotted. Pairwise differential analysis was performed between larvae from the KED or THI after receiving identical complex bacterial communities. (XLSX)

S3 Table. ID of the 22 bacterial isolates sequenced for use in the gnotobiotic assay. (XLSX)

Acknowledgments

We would like to thank Jiehua Zhou and Ruimei Yun of the UTMB insectary core. We would also like to thank Assyatou Gueye and Marieme Gueye for their assistance in field sampling. We would like to thank Dr. Noah Rose and Dr. Caroline McBride for sharing the mosquito colonies with us. We would also like to thank Dr. Scott Weaver for his assistance in establishing collaborations to facilitate the field sampling.

Author Contributions

Conceptualization: Laura B. Dickson.

Data curation: Kamil Khanipov.

Formal analysis: Anastasia Accoti, Kamil Khanipov.

Investigation: Anastasia Accoti, Laura C. Multini, Babakar Diouf, Margaret Becker, Julia Vulcan, Dianne Y. Yap, Alioune Gaye.

Resources: Massamba Sylla.

Supervision: Mawlouth Diallo.

Writing - original draft: Anastasia Accoti, Laura C. Multini, Laura B. Dickson.

Writing - review & editing: Anastasia Accoti, Laura C. Multini, Laura B. Dickson.

References

 Bhatt S., Gething P.W., Brady O.J., Messina J.P., Farlow A.W., Moyes C.L., Drake J.M., Brownstein J. S., Hoen A.G., Sankoh O., Myers M.F., George D.B., Jaenisch T., Wint G.R., Simmons C.P., Scott T. W., Farrar J.J., Hay S.I., The global distribution and burden of dengue. Nature, 2013. 496(7446): p. 504–7 https://doi.org/10.1038/nature12060 PMID: 23563266

- Wilder-Smith A., Gubler D.J., Weaver S.C., Monath T.P., Heymann D.L., Scott T.W., Epidemic arboviral diseases: priorities for research and public health. Lancet Infect Dis, 2017. 17(3): p. e101–e106 https:// doi.org/10.1016/S1473-3099(16)30518-7 PMID: 28011234
- Rocklov J., Dubrow R., Climate change: an enduring challenge for vector-borne disease prevention and control. Nat Immunol, 2020. 21(5): p. 479–483 <u>https://doi.org/10.1038/s41590-020-0648-y</u> PMID: 32313242
- Robert M.A., Christofferson R.C., Weber P.D., Wearing H.J., Temperature impacts on dengue emergence in the United States: Investigating the role of seasonality and climate change. Epidemics, 2019. 28: p. 100344 https://doi.org/10.1016/j.epidem.2019.05.003 PMID: 31175008
- Mordecai E.A., Ryan S.J., Caldwell J.M., Shah M.M., LaBeaud A.D., Climate change could shift disease burden from malaria to arboviruses in Africa. Lancet Planet Health, 2020. 4(9): p. e416–e423 https:// doi.org/10.1016/S2542-5196(20)30178-9 PMID: 32918887
- Dickson L.B., Campbell C.L., Juneja P., Jiggins F.M., Sylla M., Black W.C. Exon-Enriched Libraries Reveal Large Genic Differences Between Aedes aegypti from Senegal, West Africa, and Populations Outside Africa. G3 (Bethesda), 2017. 7(2): p. 571–582 https://doi.org/10.1534/g3.116.036053 PMID: 28007834
- Rose N.H., Sylla M., Badolo A., Lutomiah J., Ayala D., Aribodor O.B., Ibe N., Akorli J., Otoo S., Mutebi J.P., Kriete A.L., Ewing E.G., Sang R., Gloria-Soria A., Powell J.R., Baker R.E., White B.J., Crawford J. E., McBride C.S., Climate and Urbanization Drive Mosquito Preference for Humans. Curr Biol, 2020. 30(18): p. 3570–3579 e6 https://doi.org/10.1016/j.cub.2020.06.092 PMID: 32707056
- Brown J.E., McBride C.S., Johnson P., Ritchie S., Paupy C., Bossin H., Lutomiah J., Fernandez-Salas I., Ponlawat A., Cornel A.J., Black W.C., Gorrochotegui-Escalante N., Urdaneta-Marquez L., Sylla M., Slotman M., O. K., Murray K O, Walker C, Powell J R, Worldwide patterns of genetic differentiation imply multiple 'domestications' of Aedes aegypti, a major vector of human diseases. Proc Biol Sci, 2011. 278(1717): p. 2446–54 https://doi.org/10.1098/rspb.2010.2469 PMID: 21227970
- Bosio C.F., Beaty B.J., Black W.C.t. Quantitative genetics of vector competence for dengue-2 virus in Aedes aegypti. Am J Trop Med Hyg, 1998. 59(6): p. 965–70 https://doi.org/10.4269/ajtmh.1998.59.965 PMID: 9886207
- Bennett K.E., Olson K.E., Munoz Mde L., Fernandez-Salas I., Farfan-Ale J.A., Higgs S., Black W.C., Beaty B.J. Variation in vector competence for dengue 2 virus among 24 collections of Aedes aegypti from Mexico and the United States. Am J Trop Med Hyg, 2002. 67(1): p. 85–92 <u>https://doi.org/10.4269/</u> ajtmh.2002.67.85 PMID: 12363070
- Gubler D.J., Nalim S., Tan R., Saipan H., Sulianti Saroso J., Variation in susceptibility to oral infection with dengue viruses among geographic strains of Aedes aegypti. Am J Trop Med Hyg, 1979. 28(6): p. 1045–52 https://doi.org/10.4269/ajtmh.1979.28.1045 PMID: 507282
- Vazeille-Falcoz M., Mousson L., Rodhain F., Chungue E., Failloux A.B., Variation in oral susceptibility to dengue type 2 virus of populations of Aedes aegypti from the islands of Tahiti and Moorea, French Polynesia. Am J Trop Med Hyg, 1999. 60(2): p. 292–9 https://doi.org/10.4269/ajtmh.1999.60.292 PMID: 10072154
- Dickson L.B., Sanchez-Vargas I., Sylla M., Fleming K., Black W.C., Vector competence in West African Aedes aegypti Is Flavivirus species and genotype dependent. PLoS Negl Trop Dis, 2014. 8(10): p. e3153 https://doi.org/10.1371/journal.pntd.0003153 PMID: 25275366
- Lambrechts L., Quantitative genetics of Aedes aegypti vector competence for dengue viruses: towards a new paradigm? Trends Parasitol, 2011. 27(3): p. 111–4 <u>https://doi.org/10.1016/j.pt.2010.12.001</u> PMID: 21215699
- Lambrechts L., Chevillon C., Albright R.G., Thaisomboonsuk B., Richardson J.H., Jarman R.G., Scott T.W., Genetic specificity and potential for local adaptation between dengue viruses and mosquito vectors. BMC Evol Biol, 2009. 9: p. 160 https://doi.org/10.1186/1471-2148-9-160 PMID: 19589156
- Westbrook C.J., Reiskind M.H., Pesko K.N., Greene K.E., Lounibos L.P., Larval environmental temperature and the susceptibility of Aedes albopictus Skuse (Diptera: Culicidae) to Chikungunya virus. Vector Borne Zoonotic Dis, 2010. 10(3): p. 241–7 https://doi.org/10.1089/vbz.2009.0035 PMID: 19725768
- Alto B.W., Bettinardi D., Temperature and dengue virus infection in mosquitoes: independent effects on the immature and adult stages. Am J Trop Med Hyg, 2013. 88(3): p. 497–505 <u>https://doi.org/10.4269/</u> ajtmh.12-0421 PMID: 23382163
- Alto B.W., Lounibos L.P., Higgs S., Juliano S.A., Larval Competition Differentially Affects Arbovirus Infection in Aedes Mosquitoes. Ecology, 2005. 86(12): p. 3279–3288 https://doi.org/10.1890/05-0209 PMID: 19096729
- Alto B.W., Lounibos L.P., Mores C.N., Reiskind M.H., Larval competition alters susceptibility of adult Aedes mosquitoes to dengue infection. Proc Biol Sci, 2008. 275(1633): p. 463–71 <u>https://doi.org/10.1098/rspb.2007.1497</u> PMID: 18077250

- Muturi E.J., Blackshear M. Jr., Montgomery A., Temperature and density-dependent effects of larval environment on Aedes aegypti competence for an alphavirus. J Vector Ecol, 2012. 37(1): p. 154–61 https://doi.org/10.1111/j.1948-7134.2012.00212.x PMID: 22548549
- Telang A., Qayum A.A., Parker A., Sacchetta B.R., Byrnes G.R., Larval nutritional stress affects vector immune traits in adult yellow fever mosquito Aedes aegypti (Stegomyia aegypti). Med Vet Entomol, 2012. 26(3): p. 271–81 https://doi.org/10.1111/j.1365-2915.2011.00993.x PMID: 22112201
- Joy T.K., Arik A.J., Corby-Harris V., Johnson A.A., Riehle M.A., The impact of larval and adult dietary restriction on lifespan, reproduction and growth in the mosquito Aedes aegypti. Exp Gerontol, 2010. 45 (9): p. 685–90 https://doi.org/10.1016/j.exger.2010.04.009 PMID: 20451597
- Takken W., Smallegange R.C., Vigneau A.J., Johnston V., Brown M., Mordue-Luntz A.J., Billingsley P. F., Larval nutrition differentially affects adult fitness and Plasmodium development in the malaria vectors Anopheles gambiae and Anopheles stephensi. Parasit Vectors, 2013. 6(1): p. 345 https://doi.org/ 10.1186/1756-3305-6-345 PMID: 24326030
- Moller-Jacobs L.L., Murdock C.C., Thomas M.B., Capacity of mosquitoes to transmit malaria depends on larval environment. Parasit Vectors, 2014. 7: p. 593 <u>https://doi.org/10.1186/s13071-014-0593-4</u> PMID: 25496502
- Paaijmans K.P., Huijben S., Githeko A.K., Takken W., Competitive interactions between larvae of the malaria mosquitoes Anopheles arabiensis and Anopheles gambiae under semi-field conditions in western Kenya. Acta Trop, 2009. 109(2): p. 124–30 https://doi.org/10.1016/j.actatropica.2008.07.010 PMID: 18760989
- Ng'habi K.R., John B., Nkwengulila G., Knols B.G., Killeen G.F., Ferguson H.M., Effect of larval crowding on mating competitiveness of Anopheles gambiae mosquitoes. Malar J, 2005. 4: p. 49 https://doi. org/10.1186/1475-2875-4-49 PMID: 16197541
- Pfaehler O., Oulo D.O., Gouagna L.C., Githure J., Guerin P.M., Influence of soil quality in the larval habitat on development of Anopheles gambiae Giles. J Vector Ecol, 2006. 31(2): p. 400–5 <u>https://doi.org/</u> 10.3376/1081-1710(2006)31[400:iosgit]2.0.co;2 PMID: 17249359
- Okech B.A., Gouagna L.C., Yan G., Githure J.I., Beier J.C., Larval habitats of Anopheles gambiae s.s. (Diptera: Culicidae) influences vector competence to Plasmodium falciparum parasites. Malar J, 2007. 6: p. 50 https://doi.org/10.1186/1475-2875-6-50 PMID: 17470293
- Roux O., Vantaux A., Roche B., Yameogo K.B., Dabire K.R., Diabate A., Simard F., Lefevre T., Evidence for carry-over effects of predator exposure on pathogen transmission potential. Proc Biol Sci, 2015. 282(1821): p. 20152430 https://doi.org/10.1098/rspb.2015.2430 PMID: 26674956
- Minard G., Mavingui P., Moro C.V., Diversity and function of bacterial microbiota in the mosquito holobiont. Parasit Vectors, 2013. 6: p. 146 https://doi.org/10.1186/1756-3305-6-146 PMID: 23688194
- Douglas A.E., Multiorganismal insects: diversity and function of resident microorganisms. Annu Rev Entomol, 2015. 60: p. 17–34 https://doi.org/10.1146/annurev-ento-010814-020822 PMID: 25341109
- Hegde S., Rasgon J.L., Hughes G.L., The microbiome modulates arbovirus transmission in mosquitoes. Curr Opin Virol, 2015. 15: p. 97–102 <u>https://doi.org/10.1016/j.coviro.2015.08.011</u> PMID: 26363996
- Bian G., Xu Y., Lu P., Xie Y., Xi Z., The endosymbiotic bacterium Wolbachia induces resistance to dengue virus in Aedes aegypti. PLoS Pathog, 2010. 6(4): p. e1000833 <u>https://doi.org/10.1371/journal.ppat.</u> 1000833 PMID: 20368968
- Cirimotich C.M., Ramirez J.L., Dimopoulos G., Native microbiota shape insect vector competence for human pathogens. Cell Host Microbe, 2011. 10(4): p. 307–10 https://doi.org/10.1016/j.chom.2011.09. 006 PMID: 22018231
- **35.** Ramirez J.L., Short S.M., Bahia A.C., Saraiva R.G., Dong Y., Kang S., Tripathi A., Mlambo G., Dimopoulos G., Chromobacterium Csp_P reduces malaria and dengue infection in vector mosquitoes and has entomopathogenic and in vitro anti-pathogen activities. PLoS Pathog, 2014. 10(10): p. e1004398 https://doi.org/10.1371/journal.ppat.1004398 PMID: 25340821
- Ramirez J.L., Souza-Neto J., Torres Cosme R., Rovira J., Ortiz A., Pascale J.M., Dimopoulos G., Reciprocal tripartite interactions between the Aedes aegypti midgut microbiota, innate immune system and dengue virus influences vector competence. PLoS Negl Trop Dis, 2012. 6(3): p. e1561 https://doi.org/10.1371/journal.pntd.0001561 PMID: 22413032
- Dickson L.B., Jiolle D., Minard G., Moltini-Conclois I., Volant S., Ghozlane A., Bouchier C., Ayala D., Paupy C., Moro C.V., Lambrechts L., Carryover effects of larval exposure to different environmental bacteria drive adult trait variation in a mosquito vector. Sci Adv, 2017. 3(8): p. e1700585 https://doi.org/ 10.1126/sciadv.1700585 PMID: 28835919
- Giraud E., Varet H., Legendre R., Sismeiro O., Aubry F., Dabo S., Dickson L.B., Valiente Moro C., Lambrechts L., Mosquito-bacteria interactions during larval development trigger metabolic changes with

carry-over effects on adult fitness. Mol Ecol, 2022. 31(5): p. 1444–1460 <u>https://doi.org/10.1111/mec.</u> 16327 PMID: 34905257

- Ryan S.J., Carlson C.J., Mordecai E.A., Johnson L.R., Global expansion and redistribution of Aedesborne virus transmission risk with climate change. PLoS Negl Trop Dis, 2019. 13(3): p. e0007213 https://doi.org/10.1371/journal.pntd.0007213 PMID: 30921321
- Iwamura T., Guzman-Holst A., Murray K.A., Accelerating invasion potential of disease vector Aedes aegypti under climate change. Nat Commun, 2020. 11(1): p. 2130 https://doi.org/10.1038/s41467-020-16010-4 PMID: 32358588
- Lambrechts L., Fansiri T., Pongsiri A., Thaisomboonsuk B., Klungthong C., Richardson J.H., Ponlawat A., Jarman R.G., Scott T.W., Dengue-1 virus clade replacement in Thailand associated with enhanced mosquito transmission. J Virol, 2012. 86(3): p. 1853–61 <u>https://doi.org/10.1128/JVI.06458-11</u> PMID: 22130539
- Guegan M., Zouache K., Demichel C., Minard G., Tran Van V., Potier P., Mavingui P., Valiente Moro C., The mosquito holobiont: fresh insight into mosquito-microbiota interactions. Microbiome, 2018. 6(1): p. 49 https://doi.org/10.1186/s40168-018-0435-2 PMID: 29554951
- 43. Roundy C.M., Azar S.R., Rossi S.L., Huang J.H., Leal G., Yun R., Fernandez-Salas I., Vitek C.J., Paploski I.A., Kitron U., Ribeiro G.S., Hanley K.A., Weaver S.C., Vasilakis N., Variation in Aedes aegypti Mosquito Competence for Zika Virus Transmission. Emerg Infect Dis, 2017. 23(4): p. 625–632 https://doi.org/10.3201/eid2304.161484 PMID: 28287375
- MacLeod H.J., Dimopoulos G., Short S.M., Larval Diet Abundance Influences Size and Composition of the Midgut Microbiota of Aedes aegypti Mosquitoes. Front Microbiol, 2021. 12: p. 645362 https://doi. org/10.3389/fmicb.2021.645362 PMID: 34220739
- Martinson V.G., Strand M.R., Diet-Microbiota Interactions Alter Mosquito Development. Front Microbiol, 2021. 12: p. 650743 https://doi.org/10.3389/fmicb.2021.650743 PMID: 34168624
- 46. Giraud E.V., Legendre R., Sismeiro O., Aubry F., Dabo S., Dickson L.B. et al. Mosquito-bacteria interactions during larval development trigger metabolic changes with carry-over effects on adult fitness. bioRxiv, 2021 https://doi.org/10.1101/2021.05.20.444942
- Louie W., Coffey L.L., Microbial Composition in Larval Water Enhances Aedes aegypti Development but Reduces Transmissibility of Zika Virus. mSphere, 2021. 6(6): p. e0068721 <u>https://doi.org/10.1128/</u> msphere.00687-21 PMID: 34878293
- Herd C.S., Grant D.G., Lin J., Franz A.W.E., Starvation at the larval stage increases the vector competence of Aedes aegypti females for Zika virus. PLoS Negl Trop Dis, 2021. 15(11): p. e0010003 https:// doi.org/10.1371/journal.pntd.0010003 PMID: 34843483
- Short S.M., Mongodin E.F., MacLeod H.J., Talyuli O.A.C., Dimopoulos G., Amino acid metabolic signaling influences Aedes aegypti midgut microbiome variability. PLoS Negl Trop Dis, 2017. 11(7): p. e0005677 https://doi.org/10.1371/journal.pntd.0005677 PMID: 28753661
- Coon K.L., Valzania L., McKinney D.A., Vogel K.J., Brown M.R., Strand M.R., Bacteria-mediated hypoxia functions as a signal for mosquito development. Proc Natl Acad Sci U S A, 2017. 114(27): p. E5362–E5369 https://doi.org/10.1073/pnas.1702983114 PMID: 28630299
- Hyde J., Correa M.A., Hughes G.L., Steven B., Brackney D.E., Limited influence of the microbiome on the transcriptional profile of female Aedes aegypti mosquitoes. Sci Rep, 2020. 10(1): p. 10880 <u>https://</u> doi.org/10.1038/s41598-020-67811-y PMID: 32616765
- Barletta A.B., Nascimento-Silva M.C., Talyuli O.A., Oliveira J.H., Pereira L.O., Oliveira P.L., Sorgine M. H., Microbiota activates IMD pathway and limits Sindbis infection in Aedes aegypti. Parasit Vectors, 2017. 10(1): p. 103 https://doi.org/10.1186/s13071-017-2040-9 PMID: 28231846
- Gabrieli P., Caccia S., Varotto-Boccazzi I., Arnoldi I., Barbieri G., Comandatore F., Epis S., Mosquito Trilogy: Microbiota, Immunity and Pathogens, and Their Implications for the Control of Disease Transmission. Front Microbiol, 2021. 12: p. 630438 <u>https://doi.org/10.3389/fmicb.2021.630438</u> PMID: 33889137
- 54. Koh C., Islam M.N., Ye Y.H., Chotiwan N., Graham B., Belisle J.T., Kouremenos K.A., Dayalan S., Tull D.L., Klatt S., Perera R., McGraw E.A., Dengue virus dominates lipid metabolism modulations in Wolbachia-coinfected Aedes aegypti. Commun Biol, 2020. 3(1): p. 518 https://doi.org/10.1038/s42003-020-01254-z PMID: 32948809
- Barletta A.B., Alves L.R., Silva M.C., Sim S., Dimopoulos G., Liechocki S., Maya-Monteiro C.M., Sorgine M.H., Emerging role of lipid droplets in Aedes aegypti immune response against bacteria and Dengue virus. Sci Rep, 2016. 6: p. 19928 https://doi.org/10.1038/srep19928 PMID: 26887863
- 56. Vial T., Marti G., Misse D., Pompon J., Lipid Interactions Between Flaviviruses and Mosquito Vectors. Front Physiol, 2021. 12: p. 763195 https://doi.org/10.3389/fphys.2021.763195 PMID: 34899388

- Marten A.D., Tift C.T., Tree M.O., Bakke J., Conway M.J., Chronic depletion of vertebrate lipids in Aedes aegypti cells dysregulates lipid metabolism and inhibits innate immunity without altering dengue infectivity. PLoS Negl Trop Dis, 2022. 16(10): p. e0010890 https://doi.org/10.1371/journal.pntd. 0010890 PMID: 36279305
- Vial T., Tan W.L., Deharo E., Misse D., Marti G., Pompon J., Mosquito metabolomics reveal that dengue virus replication requires phospholipid reconfiguration via the remodeling cycle. Proc Natl Acad Sci U S A, 2020. 117(44): p. 27627–27636 https://doi.org/10.1073/pnas.2015095117 PMID: 33087565
- 59. Chotiwan N., Andre B.G., Sanchez-Vargas I., Islam M.N., Grabowski J.M., Hopf-Jannasch A., Gough E., Nakayasu E., Blair C.D., Belisle J.T., Hill C.A., Kuhn R.J., Perera R., Dynamic remodeling of lipids coincides with dengue virus replication in the midgut of Aedes aegypti mosquitoes. PLoS Pathog, 2018. 14(2): p. e1006853 https://doi.org/10.1371/journal.ppat.1006853 PMID: 29447265
- Caragata E.P., Rances E., O'Neill S.L., McGraw E.A., Competition for amino acids between Wolbachia and the mosquito host, Aedes aegypti. Microb Ecol, 2014. 67(1): p. 205–18 https://doi.org/10.1007/s00248-013-0339-4 PMID: 24337107
- Mitri C., Bischoff E., Belda Cuesta E., Volant S., Ghozlane A., Eiglmeier K., Holm I., Dieme C., Brito-Fravallo E., Guelbeogo W.M., Sagnon N., Riehle M.M., Vernick K.D., Leucine-Rich Immune Factor APL1 Is Associated With Specific Modulation of Enteric Microbiome Taxa in the Asian Malaria Mosquito Anopheles stephensi. Front Microbiol, 2020. 11: p. 306 <u>https://doi.org/10.3389/fmicb.2020.00306</u> PMID: 32174902
- Xiao X., Yang L., Pang X., Zhang R., Zhu Y., Wang P., Gao G., Cheng G., A Mesh-Duox pathway regulates homeostasis in the insect gut. Nat Microbiol, 2017. 2: p. 17020 <u>https://doi.org/10.1038/nmicrobiol.</u> 2017.20 PMID: 28248301
- Pang X., Xiao X., Liu Y., Zhang R., Liu J., Liu Q., Wang P., Cheng G., Mosquito C-type lectins maintain gut microbiome homeostasis. Nat Microbiol, 2016. 1: p. 16023 https://doi.org/10.1038/nmicrobiol.2016. 23 PMID: 27170846
- Zhao B., Lucas K.J., Saha T.T., Ha J., Ling L., Kokoza V.A., Roy S., Raikhel A.S., MicroRNA-275 targets sarco/endoplasmic reticulum Ca2+ adenosine triphosphatase (SERCA) to control key functions in the mosquito gut. PLoS Genet, 2017. 13(8): p. e1006943 <u>https://doi.org/10.1371/journal.pgen.</u> 1006943 PMID: 28787446
- Romoli O., Gendrin M., The tripartite interactions between the mosquito, its microbiota and Plasmodium. Parasit Vectors, 2018. 11(1): p. 200 <u>https://doi.org/10.1186/s13071-018-2784-x</u> PMID: 29558973
- Souza-Neto J.A., Powell J.R., Bonizzoni M., Aedes aegypti vector competence studies: A review. Infect Genet Evol, 2019. 67: p. 191–209 https://doi.org/10.1016/j.meegid.2018.11.009 PMID: 30465912
- Caragata E.P., Tikhe C.V., Dimopoulos G., Curious entanglements: interactions between mosquitoes, their microbiota, and arboviruses. Curr Opin Virol, 2019. 37: p. 26–36 https://doi.org/10.1016/j.coviro. 2019.05.005 PMID: 31176069
- Gao H., Cui C., Wang L., Jacobs-Lorena M., Wang S., Mosquito Microbiota and Implications for Disease Control. Trends Parasitol, 2020. 36(2): p. 98–111 https://doi.org/10.1016/j.pt.2019.12.001 PMID: 31866183
- Scolari F., Casiraghi M., Bonizzoni M., Aedes spp. and Their Microbiota: A Review. Front Microbiol, 2019. 10: p. 2036 https://doi.org/10.3389/fmicb.2019.02036 PMID: 31551973
- 70. Yin C., Sun P., Yu X., Wang P., Cheng G., Roles of Symbiotic Microorganisms in Arboviral Infection of Arthropod Vectors. Trends Parasitol, 2020. 36(7): p. 607–615 <u>https://doi.org/10.1016/j.pt.2020.04.009</u> PMID: 32386795
- Cansado-Utrilla C., Zhao S.Y., McCall P.J., Coon K.L., Hughes G.L., The microbiome and mosquito vectorial capacity: rich potential for discovery and translation. Microbiome, 2021. 9(1): p. 111 <u>https:// doi.org/10.1186/s40168-021-01073-2 PMID: 34006334</u>
- 72. Douglas A.E., The Drosophila model for microbiome research. Lab Anim (NY), 2018. 47(6): p. 157–164 https://doi.org/10.1038/s41684-018-0065-0 PMID: 29795158
- 73. Chaston J.M., Dobson A.J., Newell P.D., Douglas A.E., Host Genetic Control of the Microbiota Mediates the Drosophila Nutritional Phenotype. Appl Environ Microbiol, 2016. 82(2): p. 671–9 https://doi.org/10. 1128/AEM.03301-15 PMID: 26567306
- 74. Koch H., Schmid-Hempel P., Gut microbiota instead of host genotype drive the specificity in the interaction of a natural host-parasite system. Ecol Lett, 2012. 15(10): p. 1095–103 <u>https://doi.org/10.1111/j.</u> 1461-0248.2012.01831.x PMID: 22765311
- Severson D.W., Behura S.K., Genome Investigations of Vector Competence in Aedes aegypti to Inform Novel Arbovirus Disease Control Approaches. Insects, 2016. 7(4) <u>https://doi.org/10.3390/</u> insects7040058 PMID: 27809220

- 76. White B.J., Lawniczak M.K., Cheng C., Coulibaly M.B., Wilson M.D., Sagnon N., Costantini C., Simard F., Christophides G.K., Besansky N.J., Adaptive divergence between incipient species of Anopheles gambiae increases resistance to Plasmodium. Proc Natl Acad Sci U S A, 2011. 108(1): p. 244–9 https://doi.org/10.1073/pnas.1013648108 PMID: 21173248
- 77. Aubry F., Dabo S., Manet C., Filipovic I., Rose N.H., Miot E.F., Martynow D., Baidaliuk A., Merkling S. H., Dickson L.B., Crist A.B., Anyango V.O., Romero-Vivas C.M., Vega-Rua A., Dusfour I., Jiolle D., Paupy C., Mayanja M.N., Lutwama J.J., Kohl A., Duong V., Ponlawat A., Sylla M., Akorli J., Otoo S., Lutomiah J., Sang R., Mutebi J.P., Cao-Lormeau V.M., Jarman R.G., Diagne C.T., Faye O., Faye O., Sall A.A., McBride C.S., Montagutelli X., Rasic G., Lambrechts L., Enhanced Zika virus susceptibility of globally invasive Aedes aegypti populations. Science, 2020. 370(6519): p. 991–996 https://doi.org/10. 1126/science.abd3663 PMID: 33214283
- 78. Fansiri T., Fontaine A., Diancourt L., Caro V., Thaisomboonsuk B., Richardson J.H., Jarman R.G., Ponlawat A., Lambrechts L., Genetic mapping of specific interactions between Aedes aegypti mosquitoes and dengue viruses. PLoS Genet, 2013. 9(8): p. e1003621 https://doi.org/10.1371/journal.pgen. 1003621 PMID: 23935524
- 79. Dickson L.B., Ghozlane A., Volant S., Bouchier C., Ma L., Vega-Rua A., Dusfour I., Jiolle D., Paupy C., Mayanja M.N., Kohl A., Lutwama J.J., Duong V., Lambrechts L., Diverse laboratory colonies of Aedes aegypti harbor the same adult midgut bacterial microbiome. Parasit Vectors, 2018. 11(1): p. 207 https://doi.org/10.1186/s13071-018-2780-1 PMID: 29587819
- Kozlova E.V., Hegde S., Roundy C.M., Golovko G., Saldana M.A., Hart C.E., Anderson E.R., Hornett E. A., Khanipov K., Popov V.L., Pimenova M., Zhou Y., Fovanov Y., Weaver S.C., Routh A.L., Heinz E., Hughes G.L., Microbial interactions in the mosquito gut determine Serratia colonization and blood-feeding propensity. ISME J, 2021. 15(1): p. 93–108 https://doi.org/10.1038/s41396-020-00763-3 PMID: 32895494
- Anastasia A, Shannon Q, Julia Vulcan, Cintia Cansado-Utrilla, Anderson Enyia R, Alsing Jessicaet al Microbiome variability of mosquito lines is consistent over time and across environments. bioRxiv, 2023 https://doi.org/10.1101/2023.04.17.537119.
- Hegde S., Khanipov K., Albayrak L., Golovko G., Pimenova M., Saldana M.A., Rojas M.M., Hornett E. A., Motl G.C., Fredregill C.L., Dennett J.A., Debboun M., Fofanov Y., Hughes G.L., Microbiome Interaction Networks and Community Structure From Laboratory-Reared and Field-Collected Aedes aegypti, Aedes albopictus, and Culex quinquefasciatus Mosquito Vectors. Front Microbiol, 2018. 9: p. 2160 https://doi.org/10.3389/fmicb.2018.02160 PMID: 30250462
- Coon K.L., Hegde S., Hughes G.L., Interspecies microbiome transplantation recapitulates microbial acquisition in mosquitoes. Microbiome, 2022. 10(1): p. 58 https://doi.org/10.1186/s40168-022-01256-5 PMID: 35410630
- Wu P., Sun P., Nie K., Zhu Y., Shi M., Xiao C., Liu H., Liu Q., Zhao T., Chen X., Zhou H., Wang P., Cheng G., A Gut Commensal Bacterium Promotes Mosquito Permissiveness to Arboviruses. Cell Host Microbe, 2019. 25(1): p. 101–112 e5 https://doi.org/10.1016/j.chom.2018.11.004 PMID: 30595552
- Bai L., Wang L., Vega-Rodriguez J., Wang G., Wang S., A Gut Symbiotic Bacterium Serratia marcescens Renders Mosquito Resistance to Plasmodium Infection Through Activation of Mosquito Immune Responses. Front Microbiol, 2019. 10: p. 1580 <u>https://doi.org/10.3389/fmicb.2019.01580</u> PMID: 31379768
- Coon K.L., Vogel K.J., Brown M.R., Strand M.R., Mosquitoes rely on their gut microbiota for development. Mol Ecol, 2014. 23(11): p. 2727–39 https://doi.org/10.1111/mec.12771 PMID: 24766707
- Klindworth A., Pruesse E., Schweer T., Peplies J., Quast C., Horn M., Glockner F.O., Evaluation of general 16S ribosomal RNA gene PCR primers for classical and next-generation sequencing-based diversity studies. Nucleic Acids Res, 2013. 41(1): p. e1 https://doi.org/10.1093/nar/gks808 PMID: 22933715
- Quast C., Pruesse E., Yilmaz P., Gerken J., Schweer T., Yarza P., Peplies J., Glockner F.O., The SILVA ribosomal RNA gene database project: improved data processing and web-based tools. Nucleic Acids Res, 2013. 41(Database issue): p. D590–6 <u>https://doi.org/10.1093/nar/gks1219</u> PMID: 23193283
- Dhariwal A., Chong J., Habib S., King I.L., Agellon L.B., Xia J., MicrobiomeAnalyst: a web-based tool for comprehensive statistical, visual and meta-analysis of microbiome data. Nucleic Acids Res, 2017. 45 (W1): p. W180–W188 https://doi.org/10.1093/nar/gkx295 PMID: 28449106
- 90. Chong J., Liu P., Zhou G., Xia J., Using MicrobiomeAnalyst for comprehensive statistical, functional, and meta-analysis of microbiome data. Nat Protoc, 2020. 15(3): p. 799–821 <u>https://doi.org/10.1038/s41596-019-0264-1 PMID: 31942082</u>
- Anderson M.J., Santana-Garcon J., Measures of precision for dissimilarity-based multivariate analysis of ecological communities. Ecol Lett, 2015. 18(1): p. 66–73 https://doi.org/10.1111/ele.12385 PMID: 25438826